Massachusetts Institute of Technology Handout 0 6.857: Network and Computer Security September 5, 2002 Professor Ronald L. Rivest

Student Information

Please fill in all the information and return to the TA before the end of class. (If you cannot do this now, please return it to the course secretary ASAP). Please write legibly.

Name

Year: Course/Major/University:

E-Mail address Phone: (These are only for the use of the staff...)

credit? (circle one) yes no unsure

Need help finding a group for problem sets? yes no

Which of the following classes have you taken? are taking?

6.033 6.042/18.062 6.046/18.410 6.875/18.425 18.063

If you have not taken the prerequisites for the class, why do you think you are qualified to take this class?

Have you had any other classes you believe to be relevant?

Please summarize your background in the area of computer and network security:

Why do you want to take this class?